\documentclass[10pt,a4paper,twoside,dvips]{article}

\usepackage{vmargin,fancyheadings,multicol,multirow,ifthen,cite,
            epsfig,wrapfig,calc,dcolumn,apalike,
            setspace,
            boxedminipage,rotating,
            textcomp,
            picinpar,longtable,balance}

\usepackage{imc2010}

\begin{document}
\SetPaperBodyFont



\begin{IMCpaper}{

\title{Title of this paper}
\author{S. Sheep and T. Bear
        \thanks{Armagh Observatory, College Hill, Armagh, BT61 9DG,
                United Kingdom\\
                \texttt{sheep@arm.ac.uk}
               }
       }

\abstract{Template for a paper or a poster presented at the International
Meteor Conference held in Armagh in 2010 September.}

}

\vspace*{-3\baselineskip}
\balance



\section{Introduction}

This article presents how to make a correct paper with pictures
and tables. At the end are also examples of references.

We may make small changes to the style after we have everyone's papers, when
we are finalising the proceedings volume.  Nevertheless, this example
template paper shows the style fairly similar to what we intend.  For
example, we presently use the slightly smaller font of WGN rather than the
slightly larger font of various IMC proceedings, in case there are a large
number of proceedings papers this year (since there were a large number of
contributions at the conference).

Try not to exceed 4 pages unless you have agreed with us in advance how
many pages you need.


\section{How to make a picture}

Here is a picture (Figure \ref{chairman}).  Make sure that axis labels etc.\
in plots are big enough to be readable!

\begin{figure}[!htb]\centering
\epsfig{figure = sheep_fig1.eps, width = \columnwidth}
\caption{Photograph of one of the authors chairing the IMC.}
\label{chairman}
\end{figure}

The proceedings will be printed in black and white -- be sure your figures
look ok in greyscale.  Optionally, you can submit different versions of your
figures for the electronic version of your paper (which can be in colour) and
the printed version.


\section{How to write a table}

Here is a table prepared in LaTeX (Table \ref{tab1}).

\begin{table}[!h]
\begin{center}
\begin{tabular}{@{}cc@{}}
\hline Species & Number \\
\hline
Sheep       &   1 \\
Teddy bears & 120 \\
\hline
\end{tabular}
\end{center}
\caption{Numbers of different kinds of participants attending IMC 2010.}
\label{tab1}
\end{table}


\section{Formatting and citing references}

If you know BibTeX, please use it.  The file \texttt{sheep.bib} provides
examples of references that are books (\cite{rendtelarlt08}), journals
(\cite{imc-announce}), proceedings papers (\cite{kmo}), webpages
(\cite{vaubaillon05}) or the IMO-News mailing list (\cite{mcbeath06}).

%%% Don't need to use \cite command, may be easier just to type
%%% Rendtel \& Arlt (2008), (Rendtel \& Arlt, 2008), etc.

If you don't know BibTeX, don't worry, we can format your references.


\section{Conclusion}

The file \texttt{sheep.tex} is just for those who will use Latex to write a
paper for the IMC2010 proceedings.  If you don't use \LaTeX, don't worry,
we'll format your paper in \LaTeX; you can still look at \texttt{sheep.pdf}
for an example of what a paper will look like.


\subsection*{Acknowledgements}

Thanks to all those authors who will use this template, to ease the editors'
work on Proceedings 2010.  This example Latex file is loosely based on the
equivalent provided on the website for the 2005 Oostmalle IMC proceedings.


\nocite{*}
\bibliographystyle{imo2}
\bibliography{sheep}

\end{IMCpaper}

\end{document}
